Author’s response to reviews

Title: The effectiveness of a Value-based EMOTION-cognition-Focused educational programme to reduce diabetes-related distress in Malay adults with Type 2 diabetes (VEMOFIT): study protocol for a cluster randomised controlled trial

Authors:

Boon How Chew (chewboonhow@gmail.com)

Rimke Vos (R.C.Vos-3@umcutrecht.nl)

Sazlina Shariff Ghazali (drsazsg@gmail.com)

Nurainul Hana Shamsuddin (nurainul@upm.edu.my)

Aaron Fernandez (aaronhkl@gmail.com)

Firdaus Mukhtar (drfircbt@gmail.com)

Mastura Ismail (drmastura@moh.gov.my)

Azainorsuzila Mohd Ahad (suzilahad@hotmail.com)

Narayanan Sundram (narayanansundram@gmail.com)

Siti Zubaidah Mohd Ali (dwnsiti@yahoo.com.my)

Guy Rutten (G.E.H.M.Rutten@umcutrecht.nl)

Version: 1 Date: 03 Feb 2017

Author’s response to reviews:

Thank you for your inquiry. I have added to the manuscript on page 11 line 2-3: "This pilot study had been completed, and its outcomes had informed and were incorporated into this study protocol."